Author’s response to reviews

Title: Effects of multilevel posterior ligament dissection after spinal instrumentation on adjacent segment biomechanics as a potential risk factor for proximal junctional kyphosis: a biomechanical study.

Authors:

Tobias Lange (langet@gmx.net)
Tobias Schulte (dr.tobias.schulte@web.de)
Georg Gosheger (georg.gosheger@ukmuenster.de)
Albert Schulze Boevingloh (albert.schulzeboevingloh@ukmuenster.de)
Raul Mayr (raul.mayr@tirol-kliniken.at)
Werner Schmoelz (werner.schmoelz@tirol-kliniken.at)

Version: 1 Date: 03 Oct 2017

Author’s response to reviews:

Dear Editor,

we have changed the "Availability of data and materials" section, the author's email adress and have added the requested statement on the "non-commercial interests".

Best regards

Tobias Lange